Step 4: References

It is unacceptable to have more than 3 references from the same journal. To resolve this issue and move forward in the peer-review/publication process, please revise your reference list accordingly.

The author revised the references to meet the citation requirements.

(1) Science editor:

He choice of the references is outdated.

As this is an extremely rare complication. Therefore, the relevant literature can be found very few and very old.

(2) Company editor-in-chief:

Please provide the original figure documents.

The author has provided the original image for upload.

Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

The author has made a PPT and uploaded it. The pictures are original and marked.

Please check and confirm whether the figures are original (i.e. generated de novo by the author(s) for this paper).

The author has checked the pictures and they are all original.